Author's response to reviews

Title: Immigrant women living with HIV in Spain: A qualitative approach to encourage medical follow-up

Authors:

Anne Guionnet (aguionnetlemaitre@gmail.com)
Barbara Navaza (barbaranavaza@gmail.com)
Belén Pizarro de la Fuente (beyusqui@gmail.com)
María Jesús Pérez-Elías (mjperez90@gmail.com)
Fernando Dronda (fdronda.hrc@salud.madrid.org)
Rogelio López-Vélez (rlopezvelez.hrc@salud.madrid.org)
José A Pérez Molina (jose.perezmolina@gmail.com)

Version: 2 Date: 10 July 2014

Author's response to reviews: see over
Dear Editor,

We would like to thanks the reviewers for their pertinent comments. Please find enclosed the revised manuscript where all the comments have been considered.

- Grammar and detailed editing have been reviewed, however if you consider there is still some modifications to make, please let us know and we will be pleased to make the appropriate corrections.

- The term “researchers triangulation” has been changed for the more appropriated formulation “investigator triangulation”, this analysis method also has been explained in the text.

- We changed the concept « disclosure to others and support » which expressed two ideas for “emotional support received” and we also changed the organization of the text in this part.

- The second paragraph under gender roles has been removed.

- A new pdf for the figure 2 has been proposed.

- All the comments of Susan Seibold Simpson have been considered.

We hope that the corrections enable now the publication of the article, nevertheless if the reviewers have some new propositions we remain at your disposal for any additional modifications.

Sincerely,

Anne Guionnet